Author's response to reviews

Title: Abscess of urachal remnants presenting with acute abdomen: a case series.

Authors:

Fadl M TAZI (mustaphaahsaini@hotmail.com)
Mustapha AHSAINI (drahsaini@gmail.com)
Abdelhak Khallouk (drahsaini@hotmail.com)
Soufiane MELLAS (sou3ad_123456@hotmail.com)
Roos E Stuurman-Wieringa (roos_w@hotmail.com)
Jamal M ELFASSI (ahsainimed@hotmail.fr)
My Hassan FARIH (brahim_a_01@hotmail.com)

Version: 5 Date: 4 May 2012

Author's response to reviews:

Cover letters:
- We had given the paper to an American teacher to correct all the grammatical and spelling mistakes which we wish it is ready for the publication.
- All the modification done was marked by yellow colors to facilitate the correction.